Author’s response to reviews

Title: Multiple spindle cell hemangiomas in both lungs: a rare case report and review of the literature

Authors:

Xiao Duqing (807995437@qq.com)
Wu Zhaohong (1105730906@qq.com)
Wang Gefei (18820109621@163.com)

Version: 1 Date: 17 Apr 2019

Author’s response to reviews:

Response to Reviewers:

1. Interesting manuscript, but please edit the English and also add a follow up period.

Answer: Thank you for the comments on the paper. We have edited the English and added a follow up period in my paper. All the changes highlighted with red.

2. Although you reported that no other cases of pulmonary spindle cell hemangioma were published, this is not true. There are several papers about this rare tumor, often mimicking pulmonary metastases. Your report is not enough interesting.

Answer: We would like to thank for the relevant and useful comments on the paper. The following is a response to the comments. Firstly, we have a search in PubMed by using “spindle cell hemangiomas and lung” as a search term. The outcome is zero. Secondly, we just use “spindle cell hemangiomas” as a search term. We don’t find any case report of multiple spindle cell hemangiomas in both lungs. But we find that several papers about Pulmonary epithelioid hemangioendothelioma (PEH), often mimicking pulmonary metastases. Last but not least, whether this rare tumor is reported or not, long-term follow-up and awareness is valuable.